Author's response to reviews

Title: Macular retinal and choroidal thickness in unilateral amblyopia using swept-source optical coherence tomography

Authors:

Syunsuke Araki (s_araki@med.kawasaki-m.ac.jp)
Atsushi Miki (amiki@tc5.so-net.ne.jp)
Katsutoshi Goto (k_goto@med.kawasaki-m.ac.jp)
Tsutomu Yamashita (tsutomu-y@mw.kawasaki-m.ac.jp)
Go Takizawa (tacky@med.kawasaki-m.ac.jp)
Kazuko Haruishi (kharu@med.kawasaki-m.ac.jp)
Yoshiaki Ieki (iekky@med.kawasaki-m.ac.jp)
Junichi Kiryu (kiryu@med.kawasaki-m.ac.jp)
Kiyoshi Yaoeda (surumeoyaji800@yahoo.co.jp)

Version: 2 Date: 27 Aug 2017

Author’s response to reviews:

Comment #1:

In your Ethics approval section, please clarify whether informed consent obtained from participants was written or verbal.

Response #1:

We changed it from “informed consent” to “verbal informed consent” in Methods and Ethics approval sections.

Comment #2:

Please provide a section after References where you provide the figure legends for your figure files. For each figure, the following information should be provided:

- Figure number
- Short title of figure in sentence case (required)
- Detailed legend (optional)

Response #2:

We added the Figure legends section after References.